**Author’s response to reviews**

**Title:** A 46-year-old Chinese woman presenting with retroperitoneal follicular dendritic cell sarcoma: a case report

**Authors:**

Taize Yuan (yuantaize@gzhmu.edu.cn)
Qiuxiang Yang (673126950@qq.com)
Huanhuan Zhang (373276730@qq.com)
Jian Li (lijian@163.com)
Xiuping Zhang (xiupingzhang67@163.com)

**Version:** 4 **Date:** 28 January 2014

**Author’s response to reviews:**

We have modified the manuscript (9504136971149762) according to your requirements. The following is the details

1. We have removed the dates of confinement in the Case Presentation section.

2. We have used the description of the patient’s gender of “woman” uniformly.

3. The Ethnicity of the patient was added in the Abstract and Case presentation sections.